Author's response to reviews

Title: Crotaline Fab antivenom is effective in cases of severe North American pit viper envenomation: a integrative review

Authors:

Eric J Lavonas (eric.lavonas@rmpdc.org)
Tammi H Schaeffer (tammi.schaeffer@rmpdc.org)
Jamie Kokko (jamie.kokko@rmpdc.org)
Sara L Mlynarchek (sara.mlynarchek@rmpdc.org)
Gregory M Bogdan (greg.bodgan@rmpdc.org)

Version: 2 Date: 20 March 2009

Author's response to reviews:

Dear Editors,

As requested, we have changed how we refer to the study from a “systematic review” to an integrative review. This change was made both in the title and on page 14 in the conclusion section.

Thank you again for your consideration of this article and please let us know if you have any further questions.

Sara Mlynarchek